Author’s response to reviews

Title: An updated systematic review and meta-analysis of the prevalence of hepatitis B virus in Ethiopia

Authors:
Teshiwal Yazie (teshiwalderess@gmail.com)
Mekonnen Girma (mekomekonnen2302@gmail.com)

Version: 2 Date: 14 Sep 2019

Author’s response to reviews:

Manuscript title: An updated systematic review and meta-analysis of the prevalence of hepatitis B virus in Ethiopia
Authors: Teshiwal Deress Yazie and Mekonnen Girma Tebeje
Journal name: BMC Infectious Diseases
Manuscript ID: INFD-D-19-01380

Dear editor and reviewers, as per your comments/ suggestions, we have tried to reflect accordingly.

Editor Comments:

1. It has come to our attention that within your manuscript there is significant text overlap with other publications, particularly:
   - Dear Editor according to your suggestion we have tried to fix the possible text overlaps with other publications. If there are any please help us by sending the text overlap ithenticate snapshot then we will geminately correct it.
2. We note that the author names in the system is not the same as in the title page of the manuscript. Teshiwal Yazie vs Teshiwal Deress. Please ensure that the name of authors in the system is the same as in the title page of the manuscript.
   - Dear editor, we have tried to make author information in the system and in the manuscript similar.